Author's response to reviews

Title: Contrast associated acute kidney injury by RIFLE criteria and outcome in critically ill patients of medical and surgical intensive care units

Authors:

Myounghwa Kim (kmh2050@yuhs.ac)
Sung-Won Na (NSWKSJ@yuhs.ac)
Shin Ok Koh (SOKOH@yuhs.ac)
En Jung Kim (NATLIS@yuhs.ac)
Jin Sun Cho (CHJS0214@yuhs.ac)

Version: 6
Date: 30 January 2015

Author's response to reviews:

1. Abstract
We changed conclusions of abstract as you suggested.

2. Trial Registration
We added the date of trial registration alongside the trial registration number at the end of the manuscript as you commented.

3. Figures
We removed 2 figures, and uploaded 2 figures.